Author's response to reviews

Title: A study protocol to evaluate the relationship between outdoor air pollution and pregnancy outcomes

Authors:

Manuel C Ribeiro (manuel.ribeiro@ist.utl.pt)
Maria J Pereira (maria.pereira@ist.utl.pt)
Amilcar Soares (asoares@ist.utl.pt)
Cristina Branquinho (cmbranquinho@fc.ul.pt)
Sofia Augusto (s.augusto@fc.ul.pt)
Esteve Llop (ellv66@gmail.com)
Susana Fonseca (susanafonseca@yahoo.com)
Joaquim G Nave (joaquim.g.nave@gmail.com)
Antonio B Tavares (antonio.tavares@insa.min-saude.pt)
Carlos M Dias (carlos.das@insa.min-saude.pt)
Ana Silva (ana.silva@alentejolitoral.min-saude.pt)
Ismael Selemane (ismael@csgrandola.min-saude.pt)
Joaquin de Toro (jdetoro@cssantiago.min-saude.pt)
Mario J Santos (lince@cssantiago.min-saude.pt)
Fernanda Santos (fernanda.santos@alentejolitoral.min-saude.pt)

Version: 4 Date: 6 October 2010

Author's response to reviews: see over
“A study protocol to evaluate the relationship between outdoor air pollution and pregnancy outcomes”, Manuel C Ribeiro et al.

Manuscript ID: 4545194733978541

Dear Natalie Pafitis,
Editorial Team for BioMed Central,

In response to your e-mail sent this morning, please find enclosed the revised version of manuscript. We wish to confirm the following:

— We believe revised manuscript conforms all of the points provided in the BMC medicine journals author’s checklist for manuscript formatting;
— Typing error pointed by reviewer (page 9, 2nd paragraph) is corrected.

Regards,
On behalf of all co-authors
M C Ribeiro